Author's response to reviews

Title: Is primary care ready to take on Attention Deficit Hyperactivity Disorder?

Authors:

Ajay AK Thapar (ajay.thapar@net.nl.com)
Anita A Thapar (thapar@cardiff.ac.uk)

Version: 2 Date: 10 Apr 2002

Dear Dr Collett

We attach our revised manuscript. We have addressed the referees comments as follows.

Referee 1
1. We have corrected the reference numbering.
2. We have included the references by Hill & Taylor, 2001 and “Practice parameters for the use of...” as suggested.

Referee 2

We are pleased to note the positive comments.
1. We have corrected the reference style and misspelling.
2. Psychiatrists in learning disability have been included.
3. We have changed the range for the prevalence given the narrowing of the confidence limits found by Melzer et al, as suggested.
4. We have specifically mentioned behaviour therapy as a non pharmacological intervention.

We have modified the manuscript format according to the checklist you sent. We hope that the revised version is now acceptable.

Yours sincerely,

Ajay Thapar